Author’s response to reviews

Title: Optical puncture combined with balloon dilation PCNL vs. conventional puncture dilation PCNL for kidney stones without hydronephrosis: a retrospective study

Authors:

Mi Zhou (whateverzm@163.com)
Xiang He (hx.hzsy@163.com)
Yuelong Zhang (zhngyuelong613@gmail.com)
Weiwen Yu (hfhyww@163.com)

Version: 5 Date: 11 Nov 2019

Author’s response to reviews:

Editor Comments:

1. Funding

The role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript should be declared in this section.

2. Title page

Please separate the abstract from the title page. Please also provide email addresses of all co-authors on the title page.

3. Copyedit

Your manuscript needs to be copyedited. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service.
4. Clean copy

Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Response for all the above comments: Thank you for your comments. We have revised the manuscript accordingly